Author's response to reviews

Title: Role models in a preventive program for hand eczema among healthcare workers: a qualitative exploration of their main tasks and associated barriers and facilitators.

Authors:

Anne M Bruinewoud (annebruinewoud@hotmail.com)
Esther WC van der Meer (ew.vandermeer@vumc.nl)
Joost WJ van der Gulden (Joost.vanderGulden@radboudumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Cécile RL Boot (crl.boot@vumc.nl)

Version: 5 Date: 3 August 2015

Author's response to reviews: see over
We thank the reviewers for once again reviewing our paper. Below we will respond to the points raised:

**Reviewer 1:** No comments; the revision is OK with me. Response: Thank you.

**Reviewer 2:** The authors have done a good job and have improved the manuscript in light of the reviewers’ comments. Response: Thank you.

**Minor Essential Revisions**
It is desirable that the authors still provide a reference for the constant comparison method, for instance Glaser/Strauss 1967.

Response: We have added a reference to Glaser & Strauss (nr 23). Because of this, reference number 23 and 24 changed into nrs. 24 and 25. This was changed throughout the manuscript.


**Editorial Request:** 1. Please kindly add a statement that "All authors read and approved the final manuscript" in the Authors’ Contributions Section.

Response: The authors’ contribution section now reads:

“All authors contributed to the writing of this paper. All authors read and approved the final manuscript. AMB is the principle researcher. EWCM collected the data. Together with EWCM and CRLB, AMB was responsible for the coding of the interviews and the categorization in themes. EWCM, JWJ, JRA, and CRLB supervised the study.”